Author's response to reviews

Title: Rationale and study design of PROVHLO - a worldwide multicenter randomized controlled trial on protective ventilation during general anesthesia for open abdominal surgery

Authors:

Sabrine N T Hemmes (s.hemmes@amc.uva.nl)
Paolo Severgnini (paolo.severgnini@uninsubria.it)
Samir Jaber (s-jaber@chu-montpellier.fr)
Jaime Canet (jcanet.germanstrias@gencat.cat)
Hermann Wrigge (hermann.wrigge@medizin.uni-leipzig.de)
Michael Hiesmayr (michael.hiesmayr@meduniwien.ac.at)
edda.tschernko@meduniwien.ac.at
Markus W Hollmann (m.w.hollmann@amc.uva.nl)
Jan M Binnekade (j.m.binnekade@amc.uva.nl)
Goran Hedenstierna (goran.hedenstierna@akademiska.se)
Cristian Putensen (christian.putensen@ukb.uni-bonn.de)
Marcelo Gama de Abreu (mgabreu@aol.com)
Paolo Pelosi (ppelosi@hotmail.com)
Marcus J Schultz (m.j.schultz@amc.uva.nl)

Version: 4 Date: 15 April 2011

Author's response to reviews:

Dear Editors-in-Chief,

Thank you very much for potentially accepting our manuscript.

We have made the requested adaptations.

We hope the manuscript meets your requirements.

Yours sincerely,

Sabrine Hemmes

On behalf of,

Marcus Schultz